```
Mauna Loa Solar Observatory Observer's Log
 ______
       Wed Sep 14 16:41:31 GMT 2005
Year: 05 Doy: 257
Observer: yasukawa
WEATHER COMMENT: Wed Sep 14 16:41:41 GMT 2005
In altocumulus overcast, SE wind 10 mph, temp 42 F, RH 83%.
  end
A NEW PSPT EXABYTE IS LOADED ON KENOBI Wed Sep 14 17:08:19 GMT 2005
ECHO COMMENT: Wed Sep 14 17:08:38 GMT 2005
Time OK.
 end
WEATHER COMMENT: Wed Sep 14 17:33:34 GMT 2005
Mist from high cloudbank to east beginning to blow over station.
Closing domes.
end
Wed Sep 14 19:33:04 GMT 2005
                              MKIV
                                      Start Patrol
Wed Sep 14 19:33:08 GMT 2005
                              CHIP
                                       Start Patrol
Wed Sep 14 19:33:19 GMT 2005
                              PICS
                                       Start Patrol
WEATHER COMMENT: Wed Sep 14 19:33:07 GMT 2005
Cloud situation iffy -- moisture appears to be blowing by above station
rather than at station. Opening dome and attempting observations while
watching situation closely.
Most of high clouds to west of station.
PSPT COMMENT: Wed Sep 14 19:35:42 GMT 2005
Starting up.
end
Wed Sep 14 19:51:07 GMT 2005
                              MKIV
                                      End Patrol
Wed Sep 14 20:02:46 GMT 2005
                              MKIV
                                      Start Patrol
Wed Sep 14 20:14:52 GMT 2005
                              MKIV
                                      End Patrol
WEATHER COMMENT: Wed Sep 14 20:21:16 GMT 2005
Thick dark clouds blowing over from summit. Closing domes. Idling instruments.
end
Wed Sep 14 20:21:58 GMT 2005
                              CHIP
                                       End Patrol
Wed Sep 14 20:22:12 GMT 2005
                                       End Patrol
                              PICS
GONG COMMENT: Wed Sep 14 20:44:48 GMT 2005
Went dowm to GONG to check out the gongsters' agenda. Crew will be here to
troubleshoot and hopefully resolve the camera/data computer problem that
has plagued their system for a while now. They brought out a spare camera
and will replace the old one once they learn what is happening with the
old one in place. Their plan is to return to Tucson on Monday, but
that could be extended if the problems are not resolved.
  end
NICE PICSLIMB IMAGE: 1957
```

NICE PICSLIMB IMAGE: 1957 NICE PICSDISC IMAGE: 1957 NICE CHIP IMAGE: 1957 Wed Sep 14 23:00:29 GMT 2005 MkIV

19_33.rawmk4 19_39.rawmk4 19_44.rawmk4 20_02.rawmk4 20_08.rawmk4 19_36.rawmk4 19_41.rawmk4 19_47.rawmk4 20_05.rawmk4 20_11.rawmk4